Author's response to reviews

**Title:** Effects of shared medical appointments on quality of life and cost-effectiveness for patients with a chronic neuromuscular disease. Study protocol of a randomized controlled trial.

**Authors:**

Femke M Seesing (f.seesing@neuro.umcn.nl)
Gea Drost (g.drost@neuro.umcn.nl)
Gert-Jan van der Wilt (g.vanderwilt@ebh.umcn.nl)
Baziel GM van Engelen (b.vanengelen@neuro.umcn.nl)

**Version:** 4 **Date:** 11 August 2011

**Author's response to reviews:**

Title is identical now
Reference list is placed before figure legends and appendix